Author's response to reviews

Title: Bilateral sternoclavicular joint septic arthritis secondary to indwelling central venous catheter: a case report

Authors:

Charita Pradhan (charita74@yahoo.com)
Nicholas FS Watson (nicholas.watson@nottingham.ac.uk)
Nitin Jagasia (nitinjagasia@hotmail.com)
Ray Chari (chari.ray@sfh-tr.nhs.uk)
Jane E Patterson (Jane.Patterson@sfh-tr.nhs.uk)

Version: 5 Date: 28 April 2008

Author's response to reviews:

The final revision of the accepted version has been submitted.
The consent declaration has been moved to follow the authors contributions statement.
There are no acknowledgements.
Figure 2 a&b have been submitted as a single composite figure.
Many thanks.